Author's response to reviews

Title: Assessment and management of neurogenic claudication associated with lumbar spinal stenosis in a UK primary care musculoskeletal service: a survey of current practice among physiotherapists

Authors:

Christine M Comer (c.comer@nhs.net)
Anthony C Redmond (A.Redmond@leeds.ac.uk)
Howard A Bird (Howard.Bird@leedsth.nhs.uk)
Philip G Conaghan (p.conaghan@leeds.ac.uk)

Version: 7 Date: 23 September 2009

Author's response to reviews:

Correction of simple typographical and wording errors only